Author’s response to reviews

Title: Associations between Occupational Stress, Burnout and Well-being among Manufacturing Workers: Mediating Roles of Psychological capital and Self-esteem

Authors:

Ziyue Wang (yuewzy@foxmail.com)
Hongbo Liu (hbliu@cmu.edu.cn)
Haijian Yu (YHJ19891221@yahoo.cn)
Yanwen Wu (wuyanwen_cmu@163.com)
Shuai Chang (chang9054@163.com)
Lie Wang (liewang@cmu.edu.cn)

Version: 3 Date: 20 Sep 2017

Author’s response to reviews:

Dear Editor and reviewer,

We appreciate the reviewer very much for the constructive suggestions on our manuscript entitled “Associations between Occupational Stress, Burnout and Well-being among Manufacturing Workers: Mediating Roles of Psychological capital and Self-esteem”. We revised our manuscript according to comments. All changes are marked as green color, which we would like to submit for your kind consideration. We would like to express our great appreciation to you and reviewers for comments on our paper.

Looking forward to hearing from you.

Yours sincerely,

Lie Wang.

Response the comments of Damiano Girardi (Reviewer 1): Indeed, I believe that the description of "c" (line 267) is not completely correct, because c' reflects the relation between the independent and the dependent variables, controlling for the effects of the mediators, and not and indirect relationship.
Response: This part has been revised as the reviewer’s comments (Line 267). Thank you for your professional comments.